Author's response to reviews

Title: Primary PEComa of the Bladder Treated with Primary Excision and Adjuvant Interferon-Alpha Immunotherapy: A Case Report

Authors:

Jeremy R Parfitt (irparfit@uwo.ca)
Anthony J Bella (ABella@urology.ucsf.edu)
Bret M Wehrli (bret.wehrli@lhsc.on.ca)
Jonathan I Izawa (jonathan.izawa@lhsc.on.ca)

Version: 3 Date: 20 July 2006

Author's response to reviews:

RE: Primary PEComa of the Bladder Treated with Primary Excision and Adjuvant Interferon-Alpha Immunotherapy: A Case Report (MS: 2008168897100570)

Regarding Reviewer AP's comments:

1. The percentage of tumor cells that showed positive immunostaining has been added and the use of positive and negative immunohistochemical controls has been described in detail in the "Pathology" section.
2. All references within the text are in square brackets at the end of the sentence before the full stop. This is in accordance with the instructions for authors outlined in "BMC medicine journals - Authors' checklist for manuscript formatting" (http://www.biomedcentral.com/info/ifora/medicine_journals).

Thank you for your consideration of our article.

Sincerely,

Jeremy Parfitt